\documentclass[12pt]{article}

\setlength{\parindent}{0pt}
\everymath={\displaystyle}
\pagestyle{empty}
\newcommand{\answerspace}{\vfill}

\begin{document}

\textbf{Math 126: Calculus II}
\hfill\makebox[2.5in][r]{Name:\hrulefill}

\textbf{Quiz 10}\quad \textsl{November 21, 2000}
\hfill\makebox[2.5in][r]{Section:\hrulefill}

\bigskip

1. Determine the Taylor polynomial of lowest degree that approximates
$f(x) = e^{x}$ for $|x| < 0.2$ with an error less than $10^{-3}$.
 
\answerspace

2. Compute the integral $\int_0^x \sin(t^2)\,dt$ using Maclaurin series.

\answerspace

\end{document}


